Author's response to reviews

Title: Nitric oxide synthase 2A (NOS2A) polymorphisms are not associated with invasive pneumococcal disease

Authors:

Antony Paton (Tony.Payton@man.ac.uk)
Debbie Payne (Debbie.Payne@man.ac.uk)
Limangeni A Mankhambo (lmankhambo@gmail.com)
Daniel L Banda (dlbanda@mlw.medcol.mw)
C Anthony Hart (h.c.kelly@liv.ac.uk)
William ER Olllier (Bill.Ollier@man.ac.uk)
Enitan D Carrol (edcarrol@liv.ac.uk)

Version: 4 Date: 13 March 2009

Author's response to reviews:

Dear Editor

I have only a minor correction to make.

In the acknowledgements section it should read "The former ...... and current ...... directors provided scientific and institutional support for the study"

Thank you for your consideration

Regards

Enitan Carrol